Author's response to reviews

Title: Toxic risk for stereotactic body radiotherapy with helical tomotherapy concurrent and following with erlotinib for non-small cell lung cancer patient - case report

Authors:

Chen-Hsi Hsieh (chenci28@ms49.hinet.net)  
Hou-Tai Chang (houtai38@gmail.com)  
Shih-Chiang Lin (pigchiang@yahoo.com.tw)  
Yu-Jen Chen (chenmdphd@yahoo.com)  
Li-Ying Wang (living@ntu.edu.tw)  
Yen-Ping Hsieh (fannyhsieh@hotmail.com)  
Ngot-Swan Chong (nschong@mail.femh.org.tw)  
Pei-Wei Shueng (shueng@hotmail.com)

Version: 3 Date: 26 November 2010

Author's response to reviews: see over
Professor Melissa Norton,  
December 26, 2010

Editor-in-Chief, 

*BMC Cancer*

Dear Professor Norton:

On behalf of all authors, I appreciate the time and effort of the editor and reviewers in critiquing our work (MS: 3100979884018932). Attached is a point-by-point response to the Reviewers’ comments. We submit this manuscript for reconsideration for publication in *BMC Cancer*. Thank you for your great effort on our work.

Yours sincerely,

Pei-Wei Shueng, MD

Department of Radiation Oncology,

Far Eastern Memorial Hospital,

21, Nan-Ya South Road, Section 2 Pan-Chiao, Taipei 220, Taiwan

Fax: 886-2-8966-0906

Phone: 886-2-8966-7000 ext. 1033

E-mail: shueng@hotmail.com
Reviewer - David Huang

Comment 1: Review the wording

Response: Thank you for your comments. The paper was revised by a native English speaker.

Reviewer - Maria Werner-Wasik

Comment: The manuscript is vastly improved in merit and the authors have responded to the reviewers comments appropriately and extensively. I would still like a review of English language by a fluent English speaker to smooth out some awkward sentences.

Response: Thank you for your recommendation. We will search help for native English review and has been revised as new manuscript.

Associate Editor and Section Editor

Comment: Both our Associate Editor and Section Editor feel further consideration of your manuscript is conditional on improvement of the English used. Please ensure particular attention is paid to the abstract. You should have a native English speaking colleague help you with this, if possible, or use a commercial copyediting service.

Response: We will search help for native English review and has been revised as new
manuscript.